Author’s response to reviews

Title: Three-year follow-up of physical activity in Norwegian youth from two ethnic groups: associations with socio-demographic factors

Authors:

Åse Sagatun (aase.sagatun@r-bup.no)
Elin Kolle (Elin.Kolle@nih.no)
Sigmund A. Anderssen (sigmund.anderssen@nih.no)
Magne Thoresen (magne.thoresen@medisin.uio.no)
Anne Johanne Søgaard (Anne.Johanne.Sogaard@fhi.no)

Version: 5 Date: 21 December 2008

Author’s response to reviews:

Dear Editor

In the resubmitted manuscript I have:

1. Removed a spelling error from Norwegian Institute of Public Health on the front page.
2. On the front page the following e-mail address: Anne.Johanne.Sogaard@fhi.no, is rewritten because it didn’t have a direct link (blue colour).
3. In the conclusion of the manuscript, the numbers summarising some of the findings are removed.
4. In table 2b the third decimal in the subscripts third line is removed.

Best regards,
Åse Sagatun